



P.O. BOX 1500
YELLOWKNIFE, NT X1A 2R3

Mackenzie Valley Land
& Water Board

NT567537

File _____

JUN 10 2013

May 28, 2013

Application # MV2012L80010
Copied To The / Reg

Mr. Willard Hagen, Chair
Mackenzie Valley Land and Water Board
Box 2130, 7th Floor, 4922 - 48th Street
Yellowknife, NT X1A 2P6

Dear Mr. Hagen:

**Responses to Review Comments – Roaster Complex Deconstruction Spill
Contingency Plan**

The Giant Mine Remediation Project Team (Project Team) has been monitoring the on-line review system and public registry for comments on the Roaster Complex Deconstruction Spill Contingency Plan (SCP) submitted under Water Licence MV2012L8-0010. Correspondence from only Environment Canada was received and they confirm that they have no comments on the SCP at this time.

As no specific responses are required, we trust that no further information from the Project Team is necessary for the plan approval process to continue. If you do have any questions about the SCP, please contact the undersigned by telephone at 867-669-2425 or by email at Adrian.Paradis@aandc-aadnc.gc.ca.

Yours sincerely,

Adrian Paradis
Regulatory Manager
Giant Mine Remediation Project

Canada